Author's response to reviews

Title: Marital history, health and mortality among older men and women in England and Wales

Authors:

Emily M D Grundy (emily.grundy@lshtm.ac.uk)
Cecilia Tomassini (cecilia.tomassini@unimol.it)

Version: 4 Date: 8 September 2010

Author's response to reviews:

Further comments 08 September 2010-09-08
Formatting changes to be made:

Major revisions (we require the author to make these changes)

Email addresses: Now amended

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: Have removed full stop from ms title

Figures: Now submitted as ppt with shading rather than hatches

Typography: One or two further very minor changes

Many thanks